Author's response to reviews

Title: Personalized medicine in psychiatry: problems and promises

Authors:

Uzoezi Ozomaro (uozomaro@med.miami.edu)
Claes Wahlestedt (c wahlestedt@med.miami.edu)
Charles B Nemeroff (cnemeroff@med.miami.edu)

Version: 4 Date: 2 April 2013

Author's response to reviews: see over
Author’s Cover Letter for revised submission MS (1903161683808344)

Title: Personalized medicine in psychiatry: problems and promises

Authors: Uzoezi Ozomaro, Claes Wahlestedt and Charles B Nemeroff

Version: 3 Date: 27 March 2013

Comments: Commissioned content.
Introduction to the Revised Manuscript

This document outlines our response to the reviewers’ critiques and the editorial changes that have been made.

Critique 1:

1. “Minor essential revision: The TCF4 gene has been replicated in GWAS studies in schizophrenia and should be mentioned in the text.”
   - GWAS findings re: TCF4 are now included in the text.

Editorial Changes:

1. Please ensure the manuscript is formatted to our guidelines as follows: http://www.biomedcentral.com/bmcmed/authors/instructions/review#conclusions
   - The manuscript follows the formatting guidelines as specified.

2. Remove the contents on page 2 and 3.
   - The contents section has been removed.

3. Provide keywords as follows: http://www.biomedcentral.com/bmcmed/authors/instructions/review#formatting-keywords
   - Immediately after the abstract, the following keywords have been added: Major Depressive Disorder, Schizophrenia, personalized medicine, psychiatric hereditability, epigenetics, environmental factors, endophenotypes, pharmacogenetics, neuroimaging genetics

4. Table 1 - in the legend please provide full names for SIG and NS.
   - The legend now reflects this change.

5. Figures- Are these original or have they been reproduced. If the latter, please send us permission to reproduce the figure from the copyright holder
   - The figures are originals.